Author's response to reviews

Title: Sweet's Syndrome in a patient with Crohn's disease: a case report.

Authors:

Nadia M Mustafa (nmustafa@uic.edu)
Mark Lavizzo (mlavizzo@uiuc.edu)

Version: 9 Date: 15 March 2008

Author's response to reviews:

Dear Sir/Madam

I am resubmitting the final version of my manuscript entitled 'Sweet's syndrome in a patient with Crohn's disease: a case report' after completing the formatting requirements.

I was asked to do the following formatting before submission

Qualifications - Please remove the authors' qualifications (e.g. PhD / MD) from the manuscript.

Affiliations - Each affiliation should be written in full, in the following format: department, institute, city, and country.

The abstract should be structured Introduction, Case presentation, Conclusions.

Please move the legends to a position after the reference list.

References - The reference list should contain all authors' names, if there are 30 or below. The term 'et al.' should not be used.

Please rewrite the Consent section as follows, "Written informed consent was obtained from the patient for publication of this case report and accompanying images. A copy of the written consent is available for review by the editor-in-Chief of this journal"

Please remove the blank pages 11 and 12.

Tables, move to text - You have uploaded your tables as additional files. If you intend the tables to be visible in the manuscript, please insert the tables in the manuscript file in a position after the references. Please then delete the corresponding additional files from the submission system.

Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures.
Sincerely,
Nadia Mustafa MD

Corresponding author
Resident in Internal Medicine
University of Illinois at Urbana-Champaign

824 E Kerr Ave, # 205
Urbana, IL 61802
Phone: (217) 778-6438
Fax: (217) 244-0621
E-mail: nmustafa@uiuc.edu